ORIGINAL ARTICLE



Drivers and consequences of female reproductive competition in an egalitarian, sexually monomorphic primate

Lea Prox^{1,2} · Claudia Fichtel² · Peter M. Kappeler^{1,2}

Received: 1 November 2022 / Revised: 21 April 2023 / Accepted: 26 April 2023 / Published online: 6 May 2023 © The Author(s) 2023

Abstract

Even after the 150th anniversary of sexual selection theory, the drivers and mechanisms of female sexual selection remain poorly studied. To understand demographic circumstances favoring female-female competition, trade-offs with kin selection and interactions with male reproductive strategies, we investigated female evictions in redfronted lemurs (*Eulemur rufifrons*). Based on 24 years of demographic data of known individuals, we show that female redfronted lemurs target close female kin for forcible, permanent, and presumably lethal eviction, even though groups contain multiple unrelated males whose voluntary emigration actually mitigated the probability of future female evictions. Female eviction and male emigration were predicted by group size, but male emigration was primarily driven by a proportional increase of male rivals. Female evictions were more likely than male emigrations when there were more juvenile females in a group, but the identity of evicted females was not predicted by any intrinsic traits. While birth rates were reduced by the number of juvenile females, they were higher when there were more adult females, but variation in female lifetime reproductive success was not related to any of the predictors examined here. Thus, there seems to be a limit on female group size in this lemur species. More generally, our study demonstrates a balanced interplay between female reproductive competition, competition over group membership between both sexes, and kin selection, contributing new insights into the causes and consequences of female competition in animal societies.

Significance statement

The evolutionary causes of female competition in vertebrate societies remain poorly known. Evictions represent an extreme form of female competition because even close kin are evicted when same-sized unrelated males are theoretically also available as victims. We studied drivers and consequences of evictions in redfronted lemurs (*Eulemur rufifrons*) using 24 years of demographic data from multiple groups. We show that while voluntary male emigration mitigates the probability of future female evictions, females nonetheless appear to accept the fitness costs of evicting female kin. While group size seems to be the main driver of departures by either sex, the number of juvenile females present in groups is the key variable triggering eviction events as well as physiological responses that could be interpreted as female reproductive restraint. Our study therefore revealed that competition does trump cooperation under some circumstances in the intricate interplay between sexual selection and kin selection on females.

Keywords Sexual selection · Kin selection · Female competition · Eviction · Emigration · Lemurs

Communicated by M. A van Noordwijk

Lea Prox and Claudia Fichte contributed equally to this work.

Lea Prox LProx@dpz.eu

Extended author information available on the last page of the article

Introduction

Since Charles Darwin first presented his theory of sexual selection (Darwin 1871), male mammals have long been considered to be the more competitive and aggressive sex, whereas females were portrayed as being more passive and docile (Clutton-Brock 2007; Clutton-Brock and Huchard 2013a). Even though these modal patterns have

been confirmed in recent meta-analyses (Janicke et al. 2016, 2018), this binary view of "classical" sex roles is now outdated (Schärer et al. 2012; Davidian et al. 2022; Kappeler et al. 2022a; Fromonteil et al. 2023), and the existence of diverse mechanisms of female competition is widely appreciated today (Stockley and Bro-Jørgensen 2011; Tobias et al. 2012; Hare and Simmons 2019). Yet, compared to males, the causes and mechanisms of female intrasexual competition remain understudied.

Existing evidence suggests that the causes of intrasexual competition are more diverse for females. Whereas access to potential mates is the principle cause of male-male competition, and female competition in so-called sex-role reversed species can also directly affect mating success (Clutton-Brock 2007), females may also compete for paternal care, breeding territories, or food during the non-reproductive season. The immediate fitness consequences of winning a particular agonistic interaction are smaller for females because they do not benefit as much from obtaining an additional mating opportunity that way (Clutton-Brock et al. 2006; Clutton-Brock and Huchard 2013b). Yet, the main mechanisms of competing with same-sex conspecific over reproduction have been found to be strikingly similar for both sexes (Clutton-Brock et al. 2006; Pusey 2012; Stockley and Campbell 2013). Specifically, there are numerous examples of behavioral, pheromonal, or physiological suppression of reproductive function in same-sex rivals (West-Eberhard 1979; Clutton-Brock et al. 2008; Bell et al. 2012, 2014; French et al. 2013), and androgenic steroid hormones exhibit effects on aggressive behavior and reproductive suppression in both sexes (French et al. 2013). Moreover, male and female infanticide of unrelated infants has been interpreted as a mechanism that increases the relative reproductive success of the proponents (Lukas and Huchard 2014). Finally, members of both sexes in both singular and plural breeders may evict adolescent or subordinate rivals from their territories or groups (Cant et al. 2010; Dubuc et al. 2017; Thompson et al. 2017). The frequency of eviction is usually higher among members of the philopatric sex because natal individuals of the dispersing sex pose less of a reproductive threat. As a consequence, the intensity of competition is typically higher among members of the philopatric sex because leaving or being evicted is much more costly for them (Mattison et al. 2019).

Compared to the other competitive mechanisms, eviction appears to have a strong phylogenetic signal in mammals, because it is relatively common among social herpestids, including singular breeding meerkats (Young et al. 2006; Bell et al. 2014) and plurally breeding banded mongooses (Cant et al. 2010; Thompson et al. 2016), but notably rare among primates (Kappeler and Fichtel 2012; Baniel et al. 2018). In these mongooses, evicted females or entire matrilines may return into their natal group after being evicted. This does not seem to be the case in primates, where evicted females have rarely been observed to return to their natal groups, as also indicated by the corresponding genetic signatures (Parga et al. 2015). In lemurid primates, typically an adult female (sometimes together with her juvenile offspring) is evicted (Vick and Pereira 1989), whereas the occasional mass evictions observed in some Old World primates are better described as group fissioning (Larson et al. 2018), and many evicted juvenile howler monkeys appear to leave their natal group voluntarily (Crockett and Pope 1993). In lemurs, philopatric females evict close relatives, who are typically unable to join foreign groups and rarely manage to establish a new group, presumably making evictions of female kin often fatal and therefore a key problem in sociobiology because such costly aggression toward kin is not predicted by classic kin selection theory (Hamilton 1964; West and Gardner 2010; Abbot et al. 2011).

Specifically, this theory predicts that kinship should have a modulating effect on the intensity of female competition among group members. The individual benefits of competition among females must be balanced against the inclusive fitness costs of competing with relatives, which are inevitable because of widespread female philopatry among mammals (Clutton-Brock 2021). In cases where the costs of competition are high, females compete indeed less with neighbors when they are relatives (Lambin and Yoccoz 1998), and the intensity of aggression and risk of eviction decrease with increasing kinship in some group-living species (Clutton-Brock et al. 2010).

In addition, social relationships in an established social landscape can modulate individual's dispersal or reproductive patterns (Armansin et al. 2020), leading to an adaptation of costly competition to local variation in group size and resource availability. In smaller groups, within-group feeding competition may be reduced, but group size can also be a key determinant of success in between-group competition (Lemoine et al. 2020), so that dominants should have a selfish interest in retaining a certain number of subordinates (Kappeler et al. 2009). Thus, local variation in group size is expected to reflect variation in food availability, but also the outcome of rank-dependent female reproductive strategies (Markham et al. 2015).

Finally, in species where female dispersal is not an option, low-ranking females may improve their chances of successful reproduction by mobilizing additional energy through increased glucocorticoid secretion (Beehner and Bergman 2017), and they may reduce the risk of eviction by obtaining higher social tolerance through appeasing dominants through increased grooming (Clutton-Brock et al. 2006; Kutsukake and Clutton-Brock 2006) or by foregoing reproduction (Inzani et al. 2019).

The group-living primates of Madagascar (Lemuriformes) hold great promise for insightful comparative studies on the causes, mechanisms, and functions of female competition because they exhibit widespread female dominance and genital masculinization in combination with a lack of sexual dimorphism, indicating adaptations to female competition (Kappeler and Fichtel 2015; Davidian et al. 2022). Moreover, lemur groups are relatively small, promoting high average relatedness among the members of a single matriline (Wimmer and Kappeler 2002), and characterized by male-biased adult sex ratios (Kappeler 2017). Finally, episodic targeting aggression of group members leading to severe wounding and/or eviction is common among the Lemuridae. It also occurs in captive groups and peaks during the annual mating and birth seasons (Vick and Pereira 1989), suggesting that it is functionally tied to reproductive competition rather than to feeding competition, but this notion remains untested in the wild. Victims include closely related members of the same matriline, which are not allowed to return (Vick and Pereira 1989), and it remains unresolved why females evict female kin rather than unrelated males, who are present in much higher proportions than in other primate species (Kappeler 2000; Kappeler et al. 2009).

We studied competition and dispersal in redfronted lemurs (Eulemur rufifrons), a cat-sized species living in groups of 5-12 individuals, including multiple males and females, with a male-biased adult sex ratio (Ostner and Kappeler 2004). The females of a group belong to one philopatric matriline (Wimmer and Kappeler 2002) and neither develop pronounced dominance relationships among themselves nor with males (Ostner and Kappeler 1999). As a result, social tolerance in competitive feeding situations is relatively high (Fichtel et al. 2018). Females begin reproducing in their third year of life and give birth to a single infant - potentially every year (Kappeler et al. 2022b). Reproductive activity is tightly adjusted to pronounced predictable ecological seasonality so that infants are being born near the end of a 7-month long cool dry season. In our study population, only four evicted females have been able to successfully return to their natal group in 24 study years, and only two other individuals have been observed to establish or join a new group, suggesting that it is very unlikely for an evicted female to be accepted into a non-natal group. Males do not exhibit rank-related variation in testosterone and glucocorticoid levels, but concentrations of these hormones increase during the mating and birth seasons (Ostner et al. 2008). Thus, redfronted lemurs offer a study system characterized by a perplexing combination of high general social tolerance and relaxed feeding competition with extreme bursts of potentially lethal aggression among closely related females co-residing with multiple males.

Here, we investigated patterns and drivers of female evictions and male dispersal, using demographic data collected over 24 years. Since group size was the best predictor of female evictions in a preliminary study covering only 10 years of demographic data (Kappeler and Fichtel 2012), we were specifically interested in the interplay between female evictions and male emigrations. We investigated which aspects of group composition favor female evictions over male dispersals, whether male dispersals may buffer female evictions, and which traits characterize victims of evictions. In addition, we investigated whether proxies of female competition (number of adult and juvenile females) result in reproductive restraint and consequent fitness consequences of an eviction by investigating which characteristics predict females' birth rates, early infant survival, and lifetime reproductive success.

Methods

Study site and species

Data for this study are based on long-term census observations of a population of redfronted lemurs inhabiting a local study area of ca. 80 ha in Kirindy Forest, western Madagascar. From 1995 onwards, redfronted lemurs of up to 6 adjacent groups have been captured, subjected to standard field morphometric measurements and individually marked with microtransponders and unique nylon collars (Kappeler and Fichtel 2012). In each group, one adult female has been equipped with a radio collar to facilitate near daily censuses, during which the location, activity, and composition of each group are recorded. Both immigrations and births were usually detected within a day. In the case an individual went absent and was not seen in the group for at least the following 4 weeks, we counted this event as a "disappearance." It was not possible to record data blind because our study involved focal animals in the field.

Disappearances can occur for one of two reasons: death or "departure." We distinguished between departures by females ("evictions") and those by males ("emigrations") because the former are virtually always the result of aggression, whereas the latter are typically voluntary. Departures could be confirmed by re-sightings, as for example during the dry season, when groups from outside the study area gather at waterholes within the study area that represent the only bodies of water within several kilometers (Amoroso et al. 2019). Death could be confirmed when the remains of individuals were found following predation by fosas (Cryptoprocta ferox), the main predator of lemurs in this area. Individuals that were never re-sighted were included in the counts for "unknown." For this analysis, we used data collected between 1996 and 2020, comprising 1069 group months from 4 main study groups with a total of 257 individuals (75 females, 168 males, 14 infants disappeared before they could be sexed). Over 24 years, we recorded 173 births, 263 events of disappearances, of which 39 were confirmed evictions, and 96 confirmed emigrations, 10 deaths and 118 "unknown disappearances" (Fig. 1). In some cases, more than one individual disappeared at the same time, which led to a discrepancy between the number of events (evictions: N=29; emigrations: N=69) and disappeared individuals (evictions: N=39; emigrations: N=96).

To estimate the effects of age and group composition on departures, we extracted relevant information from censuses of months in which departures occurred, i.e., the month when individuals were last seen in a group and compared them to months without departures. For the latter, we used data from the April census or the one closest to April because offspring from the previous year or immigrant males were individually marked at this time. For each event, we determined group size, numbers of adult (> 36 months) and juvenile (6–36 months) females, adult and juvenile males, the age of individuals, and cumulative rainfall of the natural year (from July of the preceding year to June of the following year) as a proxy for habitat productivity and, hence, subsequent food availability. Because climatic data were not collected regularly during the early years of the study, we used published rainfall data from the CHIRPS data base. To estimate mean relatedness among females of a group, we calculated coefficients of maternal relatedness (assuming that sisters were sired by different fathers) based on lineages for all possible female dyads.

Statistical analyses

Patterns of female evictions and male emigrations and their interplay

First, we investigated the impact of group size and composition, age, and cumulative rainfall on the likelihood for a female to be evicted from her natal group. To this end, we compared for each year adult sex ratio, group size, group composition, and rainfall in months with evictions to a month without evictions. Second, because male emigration

Fig. 1 Summary of redfronted lemur demography. Frequencies of births, deaths, evictions, and emigrations observed in four study groups over 24 years are depicted. Evictions and emigrations are combined into departures and refer to events in which an individual (females: orange; males: blue) was seen at least once alive after leaving a group. Deaths are confirmed events (remains found) and unknown includes individuals that disappeared and were not seen again



may reduce pressures to evict related females, we repeated this analysis for male emigrations. Third, we examined which factors predict whether a female or a male left the group. Fourth, we also investigated which individual traits predict female evictions, and, fifth, whether male emigrations buffer female evictions.

Reproductive restraint and fitness consequences of evictions

We additionally investigated whether female competition manifests itself in reproductive restraint such as (6) birth rates, (7) early infant survival, and (8) life-time reproductive success. We also investigated fitness consequences of evictions by estimating whether an eviction that occurred in a given year before the birth season impacted on the probability of giving birth and whether an eviction that occurred before the birth season and 3 months after births, respectively, impacted early infant survival during the first 3 months postnatally.

Model structure

Predictors of female evictions (model 1)

To estimate the probability of female evictions, we fitted a binomial GLMM with a logit link. We set the occurrence of evictions (yes, no) as the response and included rainfall, adult sex ratio (ASR; i.e., adult sex ratio measured as proportion of adult males of all adult individuals), and group size as fixed effects, and group identity as random effect with rainfall, group size, and ASR as random slopes (Barr et al. 2013). The sample for this model encompassed 123 group-level data points including 29 events of evictions involving 39 females.

As group size had a significant effect, we conducted an exploratory analysis to determine which components of group size caused the observed effect by applying multimodel inference (Barton 2018). We used this method to deal with problems of overfitting the model by including too many predictors for the given small sample size (N=29). We created a set of 15 models containing the cumulative amount of rainfall per year together with combinations of the fixed effects of number of adult females and males as well as the number of juvenile females and males. In all models, we included group identity as random effect and rainfall within group as random slope, even if more random slopes would have been theoretically identifiable for individual models, as rain was the only variable present in all models. This approach is not ideal, but to our knowledge the question of how many degrees of freedom are absorbed by random slopes is still open (Bolker et al. 2009). Hence, this is the only option, since otherwise a joined conclusion based on AICc would not be valid. To additionally control for potential collinearity, we corrected estimates by standardizing them based on partial standard deviations (Cade 2015). For each model we determined Akaike's Information Criterion, corrected for small sample size (AICc; Queen et al. 2002). For comparisons among models, we additionally determined AIC weights for each model and averaged the estimated coefficients and their standard errors using the zero method (Nakagawa and Hauber 2011; ESM Table S1, S2, Fig. S1).

Moreover, since the model compared census data of months with and without evictions, it seemed possible that the estimate for the effect of group size and ASR was biased because of the nature of data created by limiting entries of non-eviction events to 1 month of the year, i.e., April. We therefore conducted a sensitivity analysis. To this end we randomly replaced the group size or ASR obtained for a given group in a given year by a randomly selected group size or ASR value from the same group and year and a month in which no eviction event happened. We repeated this procedure 1000 times, each time fitting the full and the null model as described above, and determined the significance and also the estimate of group size. For the other fixed effect predictor (rainfall), a corresponding assessment was not required as it was constant for a given year and group. We then compared the original estimate and *P*-value for the effect of group size with the distribution of the respective estimates obtained from the repeated random selections. This revealed the estimate of group size to be slightly biased and the significance to remain essentially unaffected (ESM Fig. S2, S3).

Predictors of male emigrations (model 2)

For male emigrations, we fitted two binomial GLMMs with a logit link. Similar to the model on female evictions, we fitted one model with male emigration (yes, no) as the response, including group size, adult sex ratio, and annual rainfall as fixed effects, and group identity as random effect including rainfall within group identity as random slope. As the sample size was larger for male emigrations (N=69)than for female evictions (N=29), we did not use a multimodel inference approach to assess which characteristics of group composition predict male emigrations because we could fit all predictors in the same model. Hence, we fitted a second model with male emigration (yes/no) as the response, number of adult and juvenile females as well as number of adult and juvenile males, and rainfall as fixed effects. As random effect, we included group identity with rainfall as random slope. For model comparison, we used the AICc criterion, considering the model with a delta AIC < -2 as the one with a better fit (ESM Table S3). Since the first model including ASR revealed a better fit, we present this model in the main text. This data set encompassed a sample size of 164 group-level census data points including 69 dispersal events, comprising 96 male dispersals.

When are female evictions more likely than male emigrations? (Model 3)

To investigate under which circumstances the departing individual was more likely to be a female or a male, we included those variables that predicted female evictions and male emigrations by including all variables that had a mean weighted estimate larger than zero in the multi-model inference in the female model (model 1) or a significant effect in the male model (model 2), i.e., rainfall, group size, ASR, number of juvenile females and adult males. Since the number of adult males correlated positively with group size (Pearson correlation: N = 135, r = 0.63, P < 0.001) and ASR (Pearson correlation: N = 135, r = 0.67, P < 0.001), we did not include the number of adult males in this model. We constructed a binomial GLMM including confirmed female evictions and male emigrations by setting sex (1 = female,0 = male) as the response and rainfall, group size, ASR, and number of juvenile females as fixed effects. We included group identity as random effect and included ASR and rainfall as random slopes. The sample for this model encompassed a total of 135 departures with four out of 33 females being evicted between two and four times and 13 out of 68 males emigrating between two and six times.

Individual traits favoring female eviction (model 4)

To estimate intrinsic factors affecting the likelihood to be evicted, we included individual characteristics of all females present during eviction events with a female being evicted or not as response and female age and presence of the mother as fixed effects, and group ID, individual ID, and event ID as random effects with female age within group as random slopes. This data set included only females that were born in the population since 1996 with 111 data points of 31 census entries and 43 individuals.

Are female evictions less likely after male emigrations? (5 permutation test)

To test whether the occurrence of a male emigration event subsequently reduced the probability of a female eviction, we used a permutation test (Adams and Anthony 1996; Manly 1997) that compared the time intervals between an emigration and a subsequent eviction. To this end, we first determined for each female eviction the time lag between the last male emigration event and the eviction. We then determined the mean time lag per group and averaged it across groups. We chose this value as a test statistic.

We then permuted, separately for each group, the temporal distributions of female evictions or male emigrations, depending on which event was more common. The permutation consisted of shuffling the time lags between events. That is, the timing of the first and last event per group remained unchanged, but the intervals between them were permuted. Consequently, the principal distribution of the time lags between consecutive events remained unchanged. We conducted a total of 10,000 permutations, each time conducting the test as described above. If female evictions were less likely after male emigrations, then the majority of permuted data sets should reveal a test statistic smaller than that of the original data. Hence, we determined the P-value as the proportion of permutations revealing a test statistic at least as large as that of the original data. To avoid a P-value being exactly 0, we included the original data as one of the permutations.

Predictors of birth rates (model 6)

To examine whether female competition resulted in reproductive restraint, we constructed two GLMMs. We fitted one model examining whether a female gave birth (yes, no) as response, female age and longevity, number of co-resident adult and juvenile females in the month of birth, annual rainfall and whether an eviction occurred in the 12 months prior to the birth (yes, no) as fixed effects and female's and group identity as random effect. We included the number of adult females, female's age and longevity as well as rainfall within group and the number of adult and juvenile females, female age, rainfall and whether an eviction occurred within female's identity as random slopes without correlations between random slopes and intercept. We included females' longevity as a control factor in the model because longerlived females might be more likely to give birth (van de Pol and Verhulst 2006; Nussey et al. 2008; Kappeler et al. 2022b). Since birth rates among primates are predicted by an inverted U-shape of both the number of adult females and females' age (Dunbar and Shultz 2021), we fitted a second model by including female age, the number of adult and juvenile females as linear and squared terms as well as longevity and rainfall as control variables (ESM Table S4). For model comparison, we used the AICc criterion, considering the model with a delta AICc < -2 as the one with a better fit. This model encompassed 218 data points for 34 females from four groups giving birth to a total of 173 infants.

Predictors of infant survival (model 7)

Because infant mortality is highest early in life, we investigated whether the likelihood of an offspring to survive until 3 months of age was influenced by our proxies of female competition. To this end, we fitted a binomial GLMM on whether an infant survived to 3 months (yes, no) as the response variable. We included female age and longevity, number of coresident adult and juvenile females, annual rainfall and whether an eviction occurred before or 3 months after birth (yes, no) as fixed effects. We included the number of adult females, female age and longevity as well as rainfall within group and the number of adult and juvenile females, female age, rainfall and whether an eviction occurred within female identity as random slopes without correlations between random slopes and intercept. This model encompassed 34 females from four groups giving birth to a total of 173 infants.

Predictors of female lifetime reproductive success (model 8)

To estimate the effects of factors related to evictions on lifetime reproductive success, we fitted another GLMM with the number of offspring born by a female over her lifetime (regardless of how long they survived) as the response and included the mean relatedness to co-resident adult females over her lifetime, mean total number of females present in the group and longevity as predictors. We included group identity as a random effect. This model was slightly overdispersed (1.26). The model comprised 26 females. We restricted our data set to females of known birth and disappearance date.

Implementation

All analyses were conducted using R (version 4.1.0, R Core Team 2019), applying the function glmer from the package "lme4" (version 1.1–21, Bates et al. 2015). We used Generalized Linear Mixed Models (GLMMs; Baayen 2008) with binomial error structure and logit link function. To ease model convergence, we centered all quantitative predictors

Fig. 2 Overview of positive and negative significant effects of predictors on responses of models 1, 2, 3, 6, and 7. Dashed arrows represent negative effects on responses; nondashed arrows represent positive effects. We only included results of the main models, omitting results of multi-model inference approach to a mean of zero and a standard deviation of one before including them into the models. We included all theoretically identifiable random slopes to avoid type I errors (Queen et al. 2002; Barr et al. 2013) except where we applied model comparisons based on AICc. We compared the resulting model to a null model, comprising all random effects and control factors included in the full model but lacking the fixed effects with a likelihood ratio test (Schielzeth and Forstmeier 2009). This full-null model serves to avoid "cryptic multiple testing" (Forstmeier and Schielzeth 2011). We obtained confidence intervals for all models by means of parametric bootstraps using the function "bootMer" of the package "lme4," applying 1000 parametric bootstraps. We checked for collinearity by determining Variance Inflation Factors (VIF, 95) for a standard linear model without random effects using the package "car" (version 3.0.11, Field 2005). To estimate model stability, we proceeded by dropping levels of the random effect one at a time from the data set and compared the obtained estimates to the estimates obtained for the full data set. All models exhibited good stability.

Results

Figure 2 presents a graphical overview of the main outcomes of the models.

Patterns of female evictions and male emigrations and their interplay

Predictors of female evictions (model 1)

In total, we have evidence of 29 eviction events involving 39 females. Evictions occurred throughout the year, but they



peaked around the brief annual mating and birth seasons (Fig. 3). The model including the baseline predictors (group size, ASR, rainfall) was overall significant (full-null model comparison: $\chi^2 = 8.79$, df = 3, P = 0.032). The probability of an eviction was higher in larger groups and in years with higher rainfall, but ASR did not predict evictions (Fig. 4, Table 1a).

As group size had a significant effect, we additionally applied a multi-model inference approach based on weighted AIC weights to determine which age/sex category drove this effect. These models (N = 15) included the number of adult females and males, juvenile females and males, and cumulative rainfall. There was one model that clearly stood out with an AIC weight of 0.57 that included the number of juvenile females, adult males and rainfall as predictors (ESM Table S2, Fig S1, N juvenile females: mean weighted estimate = 0.88, SE = 0.26, number of adult males: mean weighted estimate = 0.67, SE = 0.25, rainfall: mean weighted estimate = 0.48, SE = 0.27). The number of adult females and juvenile males seemed to have no impact on female evictions (N adult females: mean weighted estimate = 0.02, SE = 0.06, N juvenile males: mean weighted estimate = -0.00, SE = 0.05). Hence, group size best predicted evictions according to our baseline model, and this effect was driven by the number of juvenile females and the number of adult males. Since the number of adult males correlates positively with group size (Pearson correlation: N = 135, r = 0.63, P < 0.001), group size seems to predict evictions better than the number of adult males.



Fig. 4 Probability of female eviction as a function of group size. Dashed line indicates the regression line and polygons the 95% confidence intervals. Point size is relative to number of observations

Predictors of male emigration (model 2)

We observed 69 dispersal events, comprising 96 male emigrations. Emigrations occurred throughout the year with a slight increase during the mating season (Fig. 3b). Male dispersals were best predicted by group size and ASR, but not by annual rainfall (full-null model comparison: $\chi^2 = 27.98$, df = 3, P < 0.001). Males were more likely to disperse from larger groups and when the ASR was more strongly male-biased (Table 1b, Fig. 5a, b).



Fig.3 Annual distribution of rainfall and group departures across the calendar year. a) Average mean monthly rainfall, b) Total number of observed evictions and emigrations in males and females. The timing

of the annual mating season is indicated by red bars and the timing of the annual birth season is indicated by yellow bars

lable 1	Results of the mode	Is estimating drivers of	f female evictions and	l male emigrations and	their interplay
---------	---------------------	--------------------------	------------------------	------------------------	-----------------

Model		Term	Estimate	SE	Lower CI	Upper CI	Р
a) Female eviction (model 1)	$\chi^2 = 8.79,$ df = 2, P = 0.032	Intercept Rainfall ^b Group size ^b ASR	-2.78 0.52 0.91 2.23	1.97 0.26 0.33 3.29	-7.30 0.06 0.37 -4.88	1.27 1.20 1.75 9.44	a 0.049 0.026 0.497
b) Male emigration (model 2)	$\chi^2 = 27.97,$ df = 3, P < 0.001	Intercept Rainfall ^b Group size ^b ASR ^b	-0.39 0.17 0.61 0.60	0.17 0.17 0.19 0.19	-0.75 -0.17 0.26 0.24	-0.08 0.52 1.07 1.05	a 0.333 <0.001 0.001
c) When are F evictions more likely than M emigra- tions? (Model 3)	$\chi^2 = 35.56,$ df = 3, P < 0.001	Intercept Juvenile females ^b ASR Group size ^b Rainfall ^b	-1.40 1.26 -0.68 0.47 0.54	0.29 0.27 0.31 0.27 0.30	-2.55 0.85 -1.63 -0.04 0.01	-0.99 2.34 -0.13 1.27 1.50	a < 0.001 0.080 0.071 0.070
d) Individual predictors of female eviction (model 4)	$\chi^2 = 1.12,$ df = 2, P = 0.553	Intercept Female age ^b Presence of mother	-1.06 -0.25 0.14	0.45 0.36 0.68	-2.38 -1.22 -1.21	-0.29 0.49 1.64	^a 0.492 0.834

^aNot shown because of having a very limited interpretation

^bz-transformed to mean of zero and a standard deviation of 1; mean and standard deviation of the original predictors: **model 1**: rainfall: mean=888.72, SD=176.16, group size: mean=8.68, SD=2.54; **model 2**: rainfall: mean=875.59, SD=177.53, group size: mean=8.83, SD=2.53; **model 3**: JF: mean=1.20, SD=0.98, group size: mean=9.93, SD=2.29; rainfall: mean=906.34, SD=162.79; **model 4**: female age: mean=80.03, SD=67.17

When are female evictions more likely than male emigrations? (Model 3)

Fig. 6). ASR and rainfall predicted only by trend whether females were more likely to depart than males, with females being more likely to depart when the ASR was female-biased and when there was more rain (Table 1c, Fig. 6b). Group size had no effect on female departures (Table 1c).

Females were more likely than males to leave a group when there were more juvenile females (full-null model comparison: $\chi^2 = 35.56$, df = 3, P < 0.001; Table 1c,



Fig. 5 Probability of male emigration as a function of a) group size and b) adult sex ratio. Dashed lines indicate the regression lines and polygons the 95% confidence intervals. Point size is relative to number of observations

Individual traits favoring female eviction (model 4)

The model examining the effect of individual traits on the probability of female eviction was overall not significant (full-null model comparison: $\chi^2 = 1.18$, df = 2, P = 0.553, Table 1d). Female age and presence of mother did not predict which female was evicted.

Are female evictions less likely after male emigrations? (Permutation test 5)

The permutation test revealed an effect for the time interval between a male emigration and a subsequent female eviction to be longer than expected by chance (P=0.049), suggesting that male emigrations buffer the occurrence of future eviction events.

Reproductive restraint and fitness consequences of female evictions

Predictors of birth rates (model 6)

The model estimating the probability of giving birth as a function of female age and longevity, number of adult and juvenile females present, rainfall, and whether an eviction occurred before the birth season was overall significant (full-null model comparison: $\chi^2 = 12.39$, df = 3, P = 0.006). We found a positive effect of the number of adult females, and rainfall on the probability of giving birth (Fig. 7a, c;



Fig. 6 Probability of the sex of the departing individual to be female as a function of the number of juvenile females. Dashed lines indicate the regression lines and polygons the 95% confidence intervals. Point size is relative to number of observations

Table 2a). We also found a negative effect of the number of juvenile females (Fig. 7b), but no effect of female age and longevity or whether an eviction occurred before the birth season on the probability of giving birth. Hence, females were more likely to give birth when there was more rainfall and when there were more adult females, but less likely when there were more juvenile females.

Predictors of infant survival (model 7)

The model estimating variation in the proportion of offspring surviving to the age of 3 months as a function of female age and longevity, number of adult and juvenile females, rainfall, and whether an eviction occurred before or 3 months or after birth was overall significant (full-null model comparison: $\chi^2 = 10.15$, df = 2, P = 0.006). We found a significant negative effect of the number of juvenile females and by trend (P = 0.050) also a negative effect of the number of adult females on early infant survival (Table 2b, Fig. 8a, b). We did not find an effect of rainfall, female age and longevity, and whether an eviction occurred before the birth season or 3 months after birth on early infant survival.

Predictors of female lifetime reproductive success (model 8)

Longevity had a positive effect, whereas the mean relatedness to adult females and the mean number of adult females had no significant individual effects on females' lifetime reproductive success (Table 2c). Since the full-null model comparison was not significant (likelihood ratio test comparing full and null model: $\chi^2 = 1.47$, df = 2, P = 0.477), individual effects will therefore not be discussed further.

Discussion

Our analyses revealed that the intensity of female competition in redfronted lemurs was modulated by group size and particularly by the number of juvenile females in a group (model 1, 3, 6, 7), and that both males and females competed for membership in groups with apparently limited size (model 2, permutation 5). The combination of a group size of about 10 individuals (Figs. 4 and 5) and years with higher rainfall made departures from a group more likely. Who actually left was influenced by the number of resident juvenile females, with female evictions being more likely than male emigrations when there were more juvenile females in a group (model 3). The identity of evicted females was not predicted by any of their intrinsic traits, however (model 4). The likelihood of a male emigration was also predicted by group size as well as by a proportional increase of male rivals (model 2), and



Fig. 7 Probability for any adult female to give birth as a function of a) the number of adult and b) juvenile females as well as c) annual rainfall. Dashed lines indicate the regression lines and polygons the 95% confidence intervals. Point size is relative to number of observations

male emigrations buffered future female evictions (permutation test 5). Female reproductive restraint manifested in reduced birth rates when the number of juvenile females was high, but an eviction before the birth season did not reduce subsequent birth rates (model 6). Birth rates were higher when there were more adult females in a group and also in years with higher rainfall. Early infant survival in turn was lower when there were more juvenile and by trend adult females in the group (model 7). Finally, variation in lifetime reproductive success was not related to any of the predictors examined here (model 8). Hence, redfronted lemurs are clearly limited in group size and have to balance the costs and benefits of an optimal group size against the additional indirect fitness costs of evicting related females. Our study revealed a corresponding finely balanced interplay between reproductive competition among females and competition over group membership among both sexes. Hence, this is one of the first studies to reveal the importance of considering all group members when investigating causes and mechanisms of competition in one sex. Unfortunately, because evictions are so short and unpredictable events, we lack quantitative behavioral data

Model		Term	Estimate	SE	Lower CI	Upper CI	Р
a) Birth rates (model 6)	$\chi^2 = 12.39, df = 3, P = 0.006$	Intercept	1.98	0.34	1.53	3.46	a
		Rainfall ^b	0.84	0.24	0.38	1.84	0.009
		Female age ^b	-0.38	0.27	-1.23	0.26	0.168
		Eviction (yes)	-0.53	0.69	-1.91	1.21	0.465
		Adult females ^b	0.63	0.25	0.07	1.29	0.012
		Juvenile females ^b	-0.52	0.24	-1.17	-0.20	0.021
		Longevity ^b	0.28	0.30	-0.22	1.16	0.332
b) Infant survival (model 7)) $\chi^2 = 10.15, df = 2, P = 0.006$	Intercept	1.57	0.31	1.21	7.74	a
		Rainfall ^b	0.15	0.23	-0.43	0.82	0.500
		Female age	-0.39	0.30	-0.39	0.36	0.191
		Eviction (yes)	0.69	0.55	0.69	3.85	0.197
		Adult females ^b	-0.65	0.22	-1.95	-0.21	0.050
		Juvenile females ^b	-0.66	0.33	-2.62	-0.24	0.015
		Longevity	0.32	0.55	-0.52	1.35	0.197
c) Lifetime reproductive suc-	$\chi^2 = 1.47,$ df = 2,	Intercept	1.23	0.30	0.64	1.64	a
cess (model 8)		Mean R to ad. females ^b	0.14	1.14	-0.13	0.44	0.313
	P = 0.477	Mean adult females ^b	0.05	0.14	-0.21	0.35	0.742
		Longevity	0.52	0.11	0.33	0.77	0.000

Table 2 Results of the models assessing female reproductive restraint

^a Not shown because of having a very limited interpretation

^bz-transformed to mean of zero and a standard deviation of one; mean and standard deviation of the original predictors: **model 6**: rainfall: mean=882.79, SD=179.23; female age: mean=7.99, SD=4.31; AF: mean=2.55, SD=0.74, longevity: mean=13.07, SD=5.71; **model 7**: longevity: mean=10.36, SD=4.65; mean AF: mean=2.75, SD=0.46, mean R to AF: mean=0.23, SD=0.13; **model 8**: rainfall: mean=878.40, SD=182.08; AF: mean2.41, SD=0.74, JF: mean=0.91, SD=0.90

to analyze the associated patterns and dynamics that would reveal the initiators of aggression, the identity of group members that join in on the aggression, as well as general traits these individuals may share. In the present study, we



Fig. 8 Probability of an infant surviving to the age of three months as a function of the number of juvenile females. Dashed lines indicate the regression lines and polygons the 95% confidence intervals. Point size is relative to number of observations

were therefore limited to analyzing the demographic causes and consequences of this type of competition.

Group size and competition for group membership

The prevailing form of competition within groups of redfronted lemurs is competition for group membership (models 1, 2). Our main model revealed that group size is the best predictor of the departure of an individual of either sex. Importantly, effects of changes in group size occur already at small absolute differences. In contrast to many other primate species, one or two individuals joining or leaving a group can therefore cause fundamental changes in the competitive regime. Variables shaping optimal group size continue to be in the focus of recent studies in behavioral ecology (Majolo et al. 2008; Markham et al. 2015; Rudolph et al. 2019). In general, members of larger groups enjoy benefits in terms of reduced predation risk and enhanced competitiveness toward neighboring groups, but they suffer costs from greater feeding competition, consensus costs during group coordination, travel costs and parasite risk.

Lemurs live in significantly smaller groups than anthropoid primates of the same body size (Kappeler and Heymann 1996), indicating that Malagasy ecosystems are less productive than other tropical primate habitats (Dewar and Richard 2007). In fact, fruit trees in Malagasy forests tend to be smaller and less abundant and carry less nitrogen and fruit than in African forests at similar latitudes (Donati et al. 2017; Federman et al. 2017). Pronounced seasonality, relatively low food availability, and year-to-year unpredictability in rainfall may therefore be the ultimate drivers of the competitive regime to which redfronted and other groupliving lemurs have responded with a unique set of adaptations (Kappeler and Fichtel 2015).

Drivers of female evictions

If a group reached a critical size, the likelihood that either a female or a male redfronted lemur departed increased. In case of evictions, this effect was best predicted by the number of juvenile females (model 1), whereas the number of adult males, which was also inferred as a significant factor by the multi-modal inference approach, co-varied with group size in this species with male-biased adult sex ratios. We therefore assume that group size is more likely to explain evictions, but we cannot exclude the possibility that the number of males also has an impact on the likelihood of female evictions. In addition, if there were more than two juvenile females in a group, the likelihood that a female was evicted was higher than the probability that a male emigrated. There was also a tendency that female evictions were more likely than male emigrations when the ASR was more female-biased (model 3). Male emigrations, in turn, seemed to buffer the probability of future female evictions (permutation 5). Hence, female evictions are a result of a fine-grained interplay between group size and composition. The fact that the number of juvenile females best predicted evictions may indicate that adult females may take future reproductive competition into consideration, but the relative importance of behavioral and physiological mechanisms mediating this response remains obscure.

The finding that higher rainfall was positively associated with evictions appears at first glance to contradict the notion that pronounced seasonality, relatively low food availability, and year-to-year unpredictability in rainfall are ultimate drivers of the competitive regime of redfronted lemurs. However, this result indicates that evictions do not seem to be proximately triggered by acute feeding competition, as fruit availability should be positively correlated with cumulative rainfall (Dunham et al. 2018), and fruit consumption correlated positively with rainfall in sympatric Verreaux' sifakas (Propithecus verreauxi) at this study site (Koch et al. 2017). Support for this assumption can also be gleaned from the fact that evictions are common in several species of provisioned captive lemur populations (Vick and Pereira 1989; Gresse et al. 1994; Digby 1999). In fact, in one group of captive redfronted lemurs, 18 evictions occurred in just 3 years (Vick and Pereira 1989), which is a much greater rate than observed in any of our wild study groups, even when counting all disappearances as evictions.

Moreover, evictions could even be more likely when evicting females are in relatively good physical condition, i.e., in years with high cumulative rainfall and more available food resources, as the process might be costly for both aggressor and victim. In captivity, where targets of aggression have limited options to escape their predominantly female aggressors, severe wounding and even cases of death have been recorded (Vick and Pereira 1989; Gresse et al. 1994), suggesting high costs for both parties. In the wild, redfronted lemurs exhibit pronounced yearto-year variation in parasite infection intensity (Clough et al. 2010), which might be associated with variation in body condition. Hence, more fine-grained data on the links between rainfall, fruit availability and body condition, as well as on the magnitude of their temporal delays will be required to formally test this postulated link between body condition and female evictions.

Kin selection theory predicts that kinship should have a modulating effect on the intensity of female competition among group members. At large group sizes, evictions in redfronted lemurs were specifically driven by the number of juvenile females in a group, suggesting that factors intrinsically related to females are also relevant for predicting mechanisms of female competition. As the number of juvenile females increases, adult females might want to reserve limited breeding positions for their own daughters, making more distantly related females targets of eviction (Soma and Koyama 2013). However, presence of the mother did not predict which female was targeted for an eviction (model 4). Banded mongooses, in contrast, discriminate negatively against closer kin when it comes to evictions (Thompson et al. 2017). The effect of the number of juvenile females on increased female competition possibly also manifests in the form of sexual mimicry of female infants in redfronted lemurs. They change from a coloration typical for adult males to a female coloration at the age of 3 to 4 months which may protect infant females from female aggression, including female infanticide (Jolly et al. 2000; Barthold et al. 2009). Hence, evictions may serve to reduce both, immediate and future reproductive competition for perpetrators and their female offspring, respectively.

Female reproductive restraint and fitness consequences of evictions

Female competition may also manifest itself by reducing other females' birth rates or offspring survival. As the numbers of females increase, their cumulative reproductive potential also increases. Assuming that feeding competition is most intense among mothers, one might expect them to prevent others from reproducing and any resulting offspring to be less likely to survive in larger groups. However, the number of adult females in a group and higher rainfall actually had a positive effect on birth rates (model 6). The occurrence of an eviction before the birth season did not impact birth rates. In contrast, but similar to evictions themselves, the number of juvenile females in a group promoted a response of adult females that could be interpreted as reproductive restraint (model 6). Since we considered here only birth rates but not aborted pregnancies or still births, the mechanisms underlying reproductive restraint have to be investigated in future physiological studies. Nevertheless, our results indicate that female reproductive competition in redfronted lemur seems to be also implemented via prenatal mechanisms and is driven by the number of juvenile but not adult females.

Similarly, early infant survival was negatively affected by the number of juvenile (and by trend adult) females (model 7). In contrast, in banded mongooses the number of surviving offspring until weaning declined beyond a critical number of adult breeding females (Cant et al. 2013). Those results suggest that female competition also impacts population growth in some species. Independent of the number of competitors, inducing abortions and committing infanticide represent alternative mechanisms of social control over reproduction (Gilchrist 2006), and female infanticide is indeed known from group-living lemurs (Jolly et al. 2000). Thus, female competition seems to have an impact on reproductive output, also via early infant survival.

In contrast to an earlier study on reproductive senescence in this study population (Kappeler et al. 2022b), we did not find that older females were less likely to give birth, whereas female longevity did not influence birth rates in both studies. Since we included only groups for which we had at least 20 years of demographic data in this study, the contrasting results might be due to different sample sizes. Early infant survival was also not predicted by female age or longevity, suggesting that female age and longevity as potential proxies for female condition did not impact early infant survival. Interestingly, females were more likely to give birth in years with higher rainfall, but it did not predict early infant survival. Hence, future studies are required to examine how food intake by mothers may impact infant survival.

Male emigrations

Our study makes a new contribution toward a more comprehensive understanding of social competition by also considering the drivers and effects of male emigration. Male emigration also reduces group size and alleviates pressures on females to evict close kin. Male emigration was predicted by the proportion of adult males to adult females in the group (model 2). While the total number of reproducing females seems to be limited, the number of males seems to be associated with the number of mating opportunities. Both sexes most likely benefit from reducing male emigration because they are either related to the male or because the number of males in the group reduces the take-over risk by foreign males (Port et al. 2010; Port and Cant 2014). This effect may also explain their unusually high levels of social tolerance (Fichtel et al. 2018), both compared to females, but also to most anthropoid primate species (Pereira 1995). Future research should now examine the behavioral mechanisms accompanying male emigration and immigration for a more complete understanding of the proximate control of group size.

In conclusion, female competition is sensitive to demographic and ecological cues, but not to inclusive fitness costs. In contrast to an earlier analysis (Kappeler and Fichtel 2012), including males in the analyses refined our understanding of the drivers of female eviction by identifying the number of juvenile females as an important specific predictor and male emigrations as a buffer of female evictions. In addition, this approach allowed us to investigate the finetuned interplay between competition and kin selection.

Supplementary information The online version contains supplementary material available at https://doi.org/10.1007/s00265-023-03330-w.

Acknowledgements We thank the Mention Zoologie et Biodiversité Animale de l'Université d'Antananarivo, the CAFF/CORE of the Direction des Eaux et Forêts and the CNFEREF Morondava for their long-term authorization and support of this study. We are grateful to the DPZ staff at Kirindy Forest for support and companionship in the field, to Andrianjanahary Tianasoa Dieu Donné, in particular, for his role in collecting demographic data, and to Ulrike Walbaum and Henning Lahmann for curating them. We thank Roger Mundry for his time and dedication to improve and validate our statistical methods, and Maria van Noordwijk and two anonymous referees for unusually constructive comments on an earlier version of this manuscript.

Author contribution Conceptualization: LP, CF, PMK Methodology: LP, CF, PMK Investigation: CF, PMK Visualization: LP Supervision: CF, PMK Writing—original draft: LP, PMK Writing—review and editing: LP, CF, PMK.

Funding Open Access funding enabled and organized by Projekt DEAL. Deutsche Forschungsgemeinschaft, grant KA 1082/41–1.

Data availability All data are available on figshare: https://doi.org/10. 6084/m9.figshare.22673398.

Declarations

Ethical approval This study adhered to the Guidelines for the Treatment of Animals in Behavioural Research and Teaching (Animal Behaviour 2020) and the legal requirements of the country (Madagascar) in which the work was carried out. The protocol for this research was approved by the Malagasy Ministry of the Environment, Water, and Forests (066, 202/15/MEEMF/SG/DGF/DAPT/SCBT; 90, 234/16/ MEEMF/SG/DGF/DAPT/SCBT.Re; 47, 215/18/MEEMF/SG/DGF/ DAPT/SCBT.Re; 052/19/MEDD/SG/DGF/DSAP/SCB.Re).

Conflict of interest The authors declare no competing interests.

Open Access This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/.

References

- Abbot P, Abe J, Alcock J et al (2011) Inclusive fitness theory and eusociality. Nature 471:E1–E4. https://doi.org/10.1038/nature09831
- Adams DC, Anthony CD (1996) Using randomization techniques to analyse behavioural data. Anim Behav 51:733–738. https://doi. org/10.1006/anbe.1996.0077
- Amoroso CR, Kappeler PM, Fichtel C, Nunn CL (2019) Fecal contamination, parasite risk, and waterhole use by wild animals in a dry deciduous forest. Behav Ecol Sociobiol 73:153. https://doi. org/10.1007/s00265-019-2769-6
- Armansin NC, Stow AJ, Cantor M, Leu ST, Klarevas-Irby JA, Chariton AA, Farine DR (2020) Social barriers in ecological landscapes: the social resistance hypothesis. Trends Ecol Evol 35:137–148. https://doi.org/10.1016/j.tree.2019.10.001
- Baayen RH (2008) Analyzing linguistic data: a practical introduction to statistics using R. Cambridge University Press, Cambridge. https://doi.org/10.1017/S0305000909990080
- Baniel A, Cowlishaw G, Huchard E (2018) Jealous females? Female competition and reproductive suppression in a wild promiscuous primate. Phil Trans R Soc B 285:20181332. https://doi.org/10. 1098/rspb.2018.1332
- Barr DJ, Levy R, Scheepers C, Tily C (2013) Random effects structure for confirmatory hypothesis testing: Keep it maximal. J Mem Lang 68:255–278. https://doi.org/10.1016/j.jml.2012.11.001
- Barthold J, Fichtel C, Kappeler PM (2009) What is it going to be? Pattern and potential function of natal coat change in sexually dichromatic redfronted lemurs (*Eulemur fulvus rufus*). Am J Phys Anthropol 138:1–10. https://doi.org/10.1002/ajpa.20868
- Barton K (2018) MuMIn: multi-model inference. R package version 1.42.1, https://cran.r-project.org/web/packages/MuMIn/index.html
- Bates D, Mächler M, Bolker B, Walker S (2015) Fitting linear mixedeffects models using lme4. J Stat Softw 67:1–48. https://doi.org/ 10.48550/arXiv.1406.5823
- Beehner JC, Bergman TJ (2017) The next step for stress research in primates: to identify relationships between glucocorticoid secretion and fitness. Horm Behav 91:68–83. https://doi.org/10.1016/j. yhbeh.2017.03.003
- Bell MBV, Nichols HJ, Gilchrist JS, Cant MA, Hodge SJ (2012) The cost of dominance: suppressing subordinate reproduction affects the reproductive success of dominant female banded mongooses. Phil Trans R Soc B 279:619–624. https://doi.org/10.1098/rspb. 2011.1093
- Bell MBV, Cant MA, Borgeaud C, Thavarajah N, Samson J, Clutton-Brock TH (2014) Suppressing subordinate reproduction provides benefits to dominants in cooperative societies of meerkats. Nat Commun 51:4499. https://doi.org/10.1038/ncomms5499
- Bolker BM, Brooks ME, Clark CJ, Geange SW, Poulsen JR, Stevens MHH, White J-SS (2009) Generalized linear mixed models:

a practical guide for ecology and evolution. Trends Ecol Evol 24:127–135. https://doi.org/10.1016/j.tree.2008.10.008

- Cade BS (2015) Model averaging and muddled multimodel inferences. Ecology 969:2370–2382. https://doi.org/10.1890/14-1639.1
- Cant MA, Hodge SJ, Bell MBV, Gilchrist JS, Nichols HJ (2010) Reproductive control via eviction but not the threat of eviction in banded mongooses. Phil Trans R Soc B 2771691:2219–2226. https://doi. org/10.1098/rspb.2009.2097
- Cant MA, Vitikainen E, Nichols HJ (2013) Demography and social evolution of banded mongooses. Adv Stud Behav 45:407–445. https://doi.org/10.1016/B978-0-12-407186-5.00006-9
- Clough D, Heistermann M, Kappeler PM (2010) Host intrinsic determinants and potential consequences of parasite infection in freeranging red-fronted lemurs *Eulemur fulvus rufus*. Am J Phys Anthropol 142:441–452. https://doi.org/10.1002/ajpa.21243
- Clutton-Brock TH (2007) Sexual selection in males and females. Science 318:1882–1885. https://doi.org/10.1126/science.113331
- Clutton-Brock TH (2021) Social evolution in mammals. Science 373:eabc9699. https://doi.org/10.1126/science.abc9699
- Clutton-Brock TH, Huchard E (2013a) Social competition and its consequences in female mammals. J Zool 289:151–171. https://doi. org/10.1126/science.1133311
- Clutton-Brock TH, Huchard E (2013b) Social competition and selection in males and females. Phil Trans R Soc B 368:20130074. https://doi.org/10.1098/rstb.2013.0074
- Clutton-Brock TH, Hodge SJ, Flower TP (2008) Group size and the suppression of subordinate reproduction in Kalahari meerkats. Anim Behav 76:689–700. https://doi.org/10.1016/j.anbehav.2008. 03.015
- Clutton-Brock TH, Hodge SJ, Spong G, Russell G, Jordan NR, Bennett NC, Sharpe LL, Manser MB (2006) Intrasexual competition and sexual selection in cooperative mammals. Nature 444:1065–1068. https://doi.org/10.1038/nature05386
- Clutton-Brock TH, Hodge SJ, Flower TP, Spong GF, Young AJ (2010) Adaptive suppression of subordinate reproduction in cooperative mammals. Am Nat 176:664–673. https://doi.org/10.1086/656492
- Crockett CM, Pope TR (1993) Consequences of sex differences in dispersal for juvenile red howler monkeys. In: Pereira ME, Fairbanks LA (eds) Juvenile primates: life history, development, and behavior. Oxford University Press, New York, pp 104–118
- Darwin C (1871) The descent of man and selection in relation to sex. Murray, London
- Davidian E, Surbeck M, Lukas D, Kappeler PM, Huchard E (2022) The eco-evolutionary landscape of power relationships between males and females. Trends Ecol Evol 37:706–718. https://doi.org/ 10.1016/j.tree.2022.04.004
- Dewar RE, Richard AF (2007) Evolution in the hypervariable environment of Madagascar. P Natl Acad Sci USA 104:13723–13727. https://doi.org/10.1073/pnas.0704346104
- Digby LJ (1999) Targeting aggression in blue-eyed black lemurs Eulemur macaco flavifrons. Primates 40:613–617
- Donati G, Santini L, Eppley TM et al (2017) Low levels of fruit nitrogen as drivers for the evolution of Madagascar's primate communities. Sci Rep 7:14406. https://doi.org/10.1038/ s41598-017-13906-y
- Dubuc C, English S, Thavarajah N, Dantzer B, Sharp SP, Spence-Jones HC, Gaynor D, Clutton-Brock TH (2017) Increased food availability raises eviction rate in a cooperative breeding mammal. Biol Lett 13:20160961. https://doi.org/10.1098/rsbl.2016.0961
- Dunbar RI, Shultz S (2021) Social complexity and the fractal structure of group size in primate social evolution. Biol Rev 96:1889–1906. https://doi.org/10.1111/brv.12730
- Dunham AE, Razafindratsima OH, Rakotonirina P, Wright PC (2018) Fruiting phenology is linked to rainfall variability in a tropical rain forest. Biotropica 50:396–404. https://doi.org/10.1111/btp.12564

- Federman S, Sinnott-Armstrong M, Baden AL, Chapman AL, Daly AL, Richard AL, Valenta AL, Donoghue AL (2017) The paucity of frugivores in Madagascar may not be due to unpredictable temperatures or fruit resources. PLoS ONE 12:e0168943. https://doi.org/10.1371/journal.pone.0168943
- Fichtel C, Schnoell AV, Kappeler PM (2018) Measuring social tolerance: an experimental approach in two lemurid primates. Ethology 124:65–73. https://doi.org/10.1111/eth.12706
- Field A (2005) Discovering statistics using SPSS introducing statistical methods. Sage Publications, Thousand Oaks
- Forstmeier W, Schielzeth H (2011) Cryptic multiple hypotheses testing in linear models: overestimated effect sizes and the winner's curse. Behav Ecol Sociobiol 65:47–55. https://doi.org/10.1007/ s00265-010-1038-5
- French JA, Mustoe AC, Cavanaugh J, Birnie AK (2013) The influence of androgenic steroid hormones on female aggression in 'atypical'mammals. Phil Trans R Soc B 368:20130084. https:// doi.org/10.1098/rstb.2013.0084
- Fromonteil S, Marie-Orleach L, Winkler L, Janicke T (2023) Sexual selection in females and the evolution of polyandry. PLoS Biol 21:e3001916. https://doi.org/10.1371/journal.pbio.3001916
- Gilchrist JS (2006) Female eviction, abortion, and infanticide in banded mongooses *Mungos mungo*: implications for social control of reproduction and synchronized parturition. Behav Ecol 17:664–669. https://doi.org/10.1093/beheco/ark012
- Gresse M, Lacour MCI, Fornasieri I, Roeder JJ (1994) Targeting aggression in *Lemur fulvus albifrons*. In: Roeder JJ, Thierry B, Anderson J, Herrenschmidt N (eds) Current Primate Biology II. Université Louis Pasteur, Strasbourg, pp 233–239
- Hamilton WD (1964) The genetical evolution of social behaviour. J Theor Biol 7:1–52. https://doi.org/10.1016/0022-5193(64) 90039-6
- Hare RM, Simmons LW (2019) Sexual selection and its evolutionary consequences in female animals. Biol Rev 94:929–956. https://doi.org/10.1111/brv.12484
- Inzani E, Marshall HH, Thompson FJ, Kalema-Zikusoka G, Cant MA, Vitikainen MA (2019) Spontaneous abortion as a response to reproductive conflict in the banded mongoose. Biol Lett 15:20190529. https://doi.org/10.1098/rsbl.2019.0529
- Janicke T, Morrow EH (2018) Operational sex ratio predicts the opportunity and direction of sexual selection across animals. Ecol Lett 21:384–391. https://doi.org/10.1111/ele.12907
- Janicke T, Häderer IK, Lajeunesse MJ, Anthes N (2016) Darwinian sex roles confirmed across the animal kingdom. Sci Adv 2:e1500983. https://doi.org/10.1126/sciadv.1500983
- Jolly A, Caless S, Cavigelli S, Gould S, Pereira ME, Pitts A, Pride ME, Rabenandrasana ME, Walker JD, Zafison JD (2000) Infant killing, wounding and predation in *Eulemur* and *Lemur*. Int J Primatol 21:21–40. https://doi.org/10.1023/A:1005467411880
- Kappeler PM (2000) Causes and consequences of unusual sex ratios among lemurs In: Kappeler PM (ed) Primate males: causes and consequences of variation in group composition. Cambridge University Press, Cambridge, 55–63. https://doi.org/10.1086/ 343678
- Kappeler PM (2017) Sex roles and adult sex ratios: insights from mammalian biology and consequences for primate behaviour. Phil Trans R Soc B 372:20160321. https://doi.org/10.1098/rstb. 2016.0321
- Kappeler PM, Heymann EW (1996) Nonconvergence in the evolution of primate life history and socio-ecology. Biol J Linn Soc 59:297–326. https://doi.org/10.1111/j.1095-8312.1996.tb014 68.x
- Kappeler PM, Fichtel C (2012) Female reproductive competition in Eulemur rufifrons: eviction and reproductive restraint in a plurally breeding Malagasy primate. Mol Ecol 213:685–698

- Kappeler PM, Fichtel C (2015) Eco-evo-devo of the lemur syndrome: did adaptive behavioral plasticity get canalized in a large primate radiation? Front Zool 12:S15. https://doi.org/10.1186/ 1742-9994-12-S1-S15
- Kappeler PM, Mass V, Port M (2009) Even adult sex ratios in lemurs: potential costs and benefits of subordinate males in Verreaux's sifaka *Propithecus verreauxi* in the Kirindy Forest CFPF, Madagascar. Am J Phys Anthropol 140:487–497. https://doi.org/10. 1002/ajpa.21091
- Kappeler PM, Benhaiem S, Fichtel C, Fromhage L, Höner OP, Jennions MD, Kaiser S, Krüger O, Schneider JM, Tuni C, van Schaik J, Goymann W (2022a) Sex roles and sex ratios in animals. Biol Rev 98:462–480. https://doi.org/10.1111/brv.12915
- Kappeler PM, Prox L, Pethig L, Fichtel C (2022b) Reproductive senescence in two lemur lineages. Front Ecol Evol 10:894344. https:// doi.org/10.3389/fevo.2022.894344
- Koch F, Ganzhorn JU, Rothman JM, Chapman CA, Fichtel C (2017) Sex and seasonal differences in diet and nutrient intake in Verreaux's sifakas *Propithecus verreauxi*. Am J Primatol 79(4):1–10. https://doi.org/10.1002/ajp.22595
- Kutsukake N, Clutton Brock TH (2006) Social functions of allogrooming in cooperatively breeding meerkats. Anim Behav 72:59–68. https://doi.org/10.1016/j.anbehav.2006.02.016
- Lambin X, Yoccoz NG (1998) The impact of population kin-structure on nestling survival in Townsend's voles, *Microtus townsendii*. J Anim Ecol 67:1–16. https://doi.org/10.1046/j.1365-2656.1998. 00181.x
- Larson SM, Ruiz-Lambides A, Platt ML, Brent ML (2018) Social network dynamics precede a mass eviction in group-living rhesus macaques. Anim Behav 136:185–193. https://doi.org/10.1016/j. anbehav.2017.08.019
- Lemoine S, Boesch C, Preis A, Samuni L, Crockford C, Wittig RM (2020) Group dominance increases territory size and reduces neighbour pressure in wild chimpanzees. R Soc Open Sci 7:200577. https://doi.org/10.1098/rsos.200577
- Lukas DE, Huchard E (2014) The evolution of infanticide by males in mammalian societies. Science 346:841–844. 1126/ science.1257226
- Majolo B, de Bortoli VA, Schino G (2008) Costs and benefits of group living in primates: group size effects on behaviour and demography. Anim Behav 76:1235–1247. https://doi.org/10.1016/j.anbeh av.2008.06.008
- Manly BFJ (1997) Randomization, bootstrap and Monte Carlo methods in biology, 2nd edn. Chapman and Hall, London, https://doi.org/ 10.1201/9781315273075
- Markham AC, Gesquiere LR, Alberts SC, Altmann J (2015) Optimal group size in a highly social mammal. P Natl Acad Sci USA 112:14882–14887. https://doi.org/10.1073/pnas.1517794112
- Mattison SM, Shenk MK, Thompson ME, Borgerhoff Mulder M, Fortunato L (2019) The evolution of female-biased kinship in humans and other mammals. Phil Trans R Soc B 374:2019007. https://doi. org/10.1098/rstb.2019.0007
- Nakagawa S, Hauber SME (2011) Great challenges with few subjects: statistical strategies for neuroscientists. Neurosci Biobehav Rev 35:462–473. https://doi.org/10.1016/j.neubiorev.2010.06.003
- Nussey DH et al (2008) Testing for genetic trade-offs between earlyand late-life reproduction in a wild red deer population. Proc R Soc Lond B 275:745–750. https://doi.org/10.1098/rspb.2007.0986
- Ostner J, Kappeler PM (1999) Central males instead of multiple pairs in redfronted lemurs, *Eulemur fulvus rufus* Primates, Lemuridae? Anim Behav 58:1069–1078. https://doi.org/10.1006/anbe.1999. 1222
- Ostner J, Kappeler PM (2004) Male life history and the unusual adult sex ratios of redfronted lemur, *Eulemur fulvus rufus*, groups. Anim Behav 67:249–259. https://doi.org/10.1016/j.anbehav.2003.05.012

- Ostner J, Kappeler PM, Heistermann M (2008) Androgen and glucocorticoid levels reflect seasonally occurring social challenges in male redfronted lemurs *Eulemur fulvus rufus*. Behav Ecol Sociobiol 624:627–638. https://doi.org/10.1007/s00265-007-0487-y
- Parga JA, Sauther ML, Cuozzo FP, Jacky IAY, Gould L, Sussman RW, Lawler LL, Pastorini J (2015) Genetic evidence for male and female dispersal in wild *Lemur catta*. Folia Primatol 86:66–75. https://doi.org/10.1159/000369386
- Pereira ME (1995) Development and social dominance among groupliving primates. Am J Primatol 37:143–175. https://doi.org/10. 1002/ajp.1350370207
- Port M, Cant MA (2014) Reproductive competition among males in multimale groups of primates: modeling the costs and effectiveness of conflict. Int J Primatol 35:746–763. https://doi.org/10. 1007/s10764-013-9744-2
- Port M, Johnstone RA, Kappeler PM (2010) Costs and benefits of multi-male associations in redfronted lemurs *Eulemur fulvus rufus*. Biol Lett 6:620–622. https://doi.org/10.1098/rsbl.2010. 0091
- Pusey A (2012) Magnitude and sources of variation in female reproductive performance. In: Mitani JC, Call J, Kappeler MP, Palombit RA, Silk JB (eds) The evolution of primate societies. University of Chicago Press, Chicago, pp 343–366
- Queen JP, Quinn GP, Keough MJ (2002) Experimental design and data analysis for biologists. Cambridge University Press, Cambridge. 1017/CBO9780511806384
- Rudolph K, Fichtel C, Schneider D, Heistermann M, Koch F, Daniel R, Kappeler PM (2019) One size fits all? Relationships among group size, health, and ecology indicate a lack of an optimal group size in a wild lemur population. Behav Ecol Sociobiol 73:132. https:// doi.org/10.1007/s00265-019-2746-0
- Schärer L, Rowe L, Arnqvist G (2012) Anisogamy, chance and the evolution of sex roles. Trends Ecol Evol 27:260–264. https://doi. org/10.1016/j.tree.2011.12.006
- Schielzeth H, Forstmeier W (2009) Conclusions beyond support: overconfident estimates in mixed models. Behav Ecol 20:416–420. https://doi.org/10.1093/beheco/arn145
- Soma T, Koyama N (2013) Eviction and troop reconstruction in a single matriline of ring-tailed lemurs *Lemur catta*: what happened when "grandmother" died? In: Masters J, Gamba M, Génin F (eds) Leaping ahead: advances in prosimian biology. Springer New York, New York, 137–146. https://doi.org/10.1007/978-1-4614-4511-1_16

Authors and Affiliations

Lea Prox^{1,2} · Claudia Fichtel² · Peter M. Kappeler^{1,2}

¹ Department of Sociobiology/Anthropology, University of Göttingen, Kellnerweg 6, 30077 Göttingen, Germany

- Stockley P, Bro-Jørgensen J (2011) Female competition and its evolutionary consequences in mammals. Biol Rev 86:341–366. https:// doi.org/10.1111/j.1469-185X.2010.00149.x
- Stockley P, Campbell A (2013) Female competition and aggression: interdisciplinary perspectives. Phil Trans R Soc B 368:20130073. https://doi.org/10.1098/rstb.2013.0073
- Thompson FJ, Marshall HH, Sanderson JL, Vitikainen EIK, Nichols HJ, Gilchrist JS, Young AJ, Hodge SJ (2016) Reproductive competition triggers mass eviction in cooperative banded mongooses. Phil Trans R Soc B 283:20152607. https://doi.org/10.1098/rspb. 2015.2607
- Thompson FJ, Cant MA, Marshall HH et al (2017) Explaining negative kin discrimination in a cooperative mammal society. P Natl Acad Sci USA 11420:5207–5212. https://doi.org/10.1073/pnas. 1612235114
- Tobias JA, Montgomerie R, Lyon BE (2012) The evolution of female ornaments and weaponry: social selection, sexual selection and ecological competition. Phil Trans R Soc B 367:2274–2293. https://doi.org/10.1098/rstb.2011.0280
- van de Pol M, Verhulst S (2006) Age-dependent traits: a new statistical model to separate within-and between-individual effects. Am Nat 167:766–773. https://doi.org/10.1086/503331
- Vick LG, Pereira ME (1989) Episodic targeting aggression and the histories of *Lemur* social groups. Behav Ecol Sociobiol 25:3–12. https://doi.org/10.1007/BF00299705
- West SA, Gardner A (2010) Altruism, spite, and greenbeards. Science 327:1341–1344. https://doi.org/10.1126/science.1178332
- West-Eberhard MJ (1979) Sexual selection, social competition, and evolution. Proc Am Philos Soc 123:222–234. https://doi.org/10. 1098/rstb.2011.0278
- Wimmer B, Kappeler PM (2002) The effects of sexual selection and life history on the genetic structure of redfronted lemur, *Eulemur fulvus rufus*, groups. Anim Behav 64:557–568. https://doi.org/10. 1006/anbe.2002.4003
- Young AJ, Carlson AA, Monfort SL, Russell AF, Bennett NC, Clutton-Brock TH (2006) Stress and the suppression of subordinate reproduction in cooperatively breeding meerkats. P Natl Acad Sci USA 103:12005–12010 10.1073pnas.0510038103

Publisher's Note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

² Behavioral Ecology & Sociobiology Unit, Leibniz Institute for Primate Research, Göttingen, Germany